 Vc/13-Jun-05
Week 23  starting 6 June 05
SM18 Operation Meeting (Tuesdays)

 Bench status  on Monday  06 June 05 @ 6/6/2005 1:46:37 PM
	5
	A1
	MBAL3101
	PT 5.2 1.5 KA Quench HF
	1.90

	3
	A2
	MBAR1141
	PT 12 Warm Up
	54.25

	9
	B1
	SSS135
	PT 14 Magnet MAPS
	296.34

	4
	B2
	SSS277
	PT 6.4 Training 4
	4.16

	8
	C1
	MBBL3124
	Prep 1 Lyre Test
	295.18

	1
	C2
	MBAR3301
	PT 12 Warm Up
	287.94

	10
	D1
	[image: image1.png]



	[image: image2.png]



	296.34

	11
	D2
	SSS613
	PT 14 Magnet MAPS
	296.88

	12
	E1
	MBBR3356
	ICS 6 Disconnect Magnet WP07
	299.56

	7
	E2
	MBBL3096
	ICS 3 Traveller after warm test
	297.68

	6
	F1
	MBAL1229
	Prep 5 Cool Down
	97

	2
	F2
	MBBL3134
	PT 12 Warm Up
	176.91


Coordinator this week: Nikolay; MTM support on call :   16-3602 
17  complete Magnet Tests cycles   ( 3147  ROBTC )
12  dipoles( 2 with MM )+ 5  SSS( 1  with MM) +  0 SSSS (with MM)
   Disconnected or ROBTC after tests last week ( Sat Midnight to Sat midnight) 
· Situation From Cryogenics:  J-P Lamboy or ACR Rep.
· Situation on Repair :   3 silver shoes done , 9 to go ?
· General Points 
· AOB

	


Tests 

	Magnets under Test from 29-05-2005 to 04-06-2005 

	-
	Bench
	Magnet
	Shafts
	+
	Start Date
	Connect
Time
	Cooling
Time
	Cold Test
Time
	Warming
Time
	Disconnect
Time
	End Date
	Total
	Signed for
	
	

	1
	TBA1
	MBAR3298 
	[image: image3.png]



	


	30-05-2005 
	31 h.
	21 h.
	22 h.
	15 h.
	13 h.
	04-06-2005 
	102 h.
	Stripping
	
	

	2
	TBA1
	MBAR3319 
	[image: image5.png]



	


	26-05-2005 
	26 h.
	24 h.
	39 h.
	15 h.
	9 h.
	30-05-2005 
	113 h.
	Stripping
	
	

	3
	TBA2
	MBAR3344 
	[image: image7.png]



	


	27-05-2005 
	26 h.
	36 h.
	75 h.
	15 h.
	10 h.
	03-06-2005 
	162 h.
	Stripping
	
	

	4
	TBB1
	SSS178 
	[image: image9.png]



	


	27-05-2005 
	39 h.
	13 h.
	14 h.
	9 h.
	8 h.
	30-05-2005 
	83 h.
	Stripping
	
	

	5
	TBB1
	SSS194 
	[image: image11.png]



	


	31-05-2005 
	31 h.
	10 h.
	18 h.
	7 h.
	6 h.
	02-06-2005 
	72 h.
	Stripping
	
	

	6
	TBB2
	SSS172 
	[image: image13.png]



	


	31-05-2005 
	26 h.
	14 h.
	20 h.
	9 h.
	12 h.
	04-06-2005 
	81 h.
	Stripping
	
	

	7
	TBB2
	SSS197 
	[image: image15.png]



	


	27-05-2005 
	24 h.
	23 h.
	16 h.
	11 h.
	13 h.
	31-05-2005 
	87 h.
	Stripping
	
	

	8
	TBC1
	MBBL2179 
	[image: image17.png]



	


	28-05-2005 
	17 h.
	32 h.
	20 h.
	14 h.
	7 h.
	01-06-2005 
	90 h.
	Stand By
	
	

	9
	TBC2
	MBBR3337 
	[image: image19.png]



	


	26-05-2005 
	25 h.
	21 h.
	21 h.
	13 h.
	9 h.
	30-05-2005 
	89 h.
	Stripping
	
	

	10
	TBC2
	MBBR3338 
	[image: image21.png]



	


	30-05-2005 
	21 h.
	25 h.
	16 h.
	14 h.
	13 h.
	03-06-2005 
	89 h.
	Stripping
	
	

	11
	TBD1
	SSS156 
	[image: image23.png]



	


	20-05-2005 
	42 h.
	11 h.
	160 h.
	39 h.
	8 h.
	31-05-2005 
	260 h.
	Stripping
	
	

	12
	TBE1
	MBAR3346 
	[image: image25.png]



	


	29-05-2005 
	23 h.
	25 h.
	15 h.
	14 h.
	7 h.
	02-06-2005 
	84 h.
	Stripping
	
	

	13
	TBE1
	MBBL2185 
	[image: image27.png]



	


	23-05-2005 
	29 h.
	35 h.
	50 h.
	16 h.
	10 h.
	29-05-2005 
	140 h.
	Stripping
	
	

	14
	TBE2
	MBAR3327 
	[image: image29.png]



	


	28-05-2005 
	23 h.
	25 h.
	22 h.
	14 h.
	11 h.
	01-06-2005 
	95 h.
	Stripping
	
	

	15
	TBF1
	MBAL3147 
	[image: image31.png]



	


	28-05-2005 
	0 h.
	25 h.
	11 h.
	14 h.
	13 h.
	31-05-2005 
	63 h.
	Stripping
	
	

	16
	TBF1
	MBAR3351 
	[image: image33.png]



	


	31-05-2005 
	31 h.
	27 h.
	18 h.
	13 h.
	11 h.
	04-06-2005 
	100 h.
	Stripping
	
	

	17
	TBF2
	MBAR3175 
	[image: image35.png]



	


	24-05-2005 
	36 h.
	43 h.
	27 h.
	16 h.
	9 h.
	29-05-2005 
	131 h.
	Stripping
	
	


Quench Information
 

	-
	Magnet
	Q1
	Q2
	Q3
	Q4
	Q5
	Q6
	Q7
	Q8
	Q9
	Q10
	Q11
	Q12
	Q13

	1
	MBAR3298
	11568.3 A.
(D2_U)
	12017 A.
(D1_U)
	12378 A.
(D1-L)
	[image: image37.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	2
	MBAR3319
	9338 A.
(D2-L-S01)
	11245 A.
(D1-U-S11)
	0 A.
(No Data)
	12850 A.
[image: image38.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	3
	MBAR3344
	12850 A.
[image: image39.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	4
	SSS178
	12613 A.
(QI _34)
	12850 A.
[image: image40.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	5
	SSS194
	12239 A.
(QE-12)
	12340 A.
(QE-34)
	12462 A.
(QE-34)
	[image: image41.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	6
	SSS172
	10511 A.
(QE_12)
	11777 A.
(QE_12)
	12280 A.
(QE_12)
	12776 A.
(QI-34)
	[image: image42.png]



	-
	-
	-
	-
	-
	-
	-
	-

	7
	SSS197
	11932 A.
(QI-34)
	12109 A.
(QI-34)
	12850 A.
(QI-34)
	[image: image43.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	8
	MBBL2179
	11421 A.
(D1 - U)
	12470 A.
(D2-U)
	12581 A.
(D2-L)
	[image: image44.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	9
	MBBR3337
	10455 A.
(D1-U)
	12849 A.
(No Data)
	11998.6 A.
(D1-U)
	12850 A.
[image: image45.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	10
	MBBR3338
	12180 A.
(D2 - U)
	12711 A.
(D1-L)
	12850 A.
[image: image46.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	11
	SSS156
	11916.5 A.
(QI-12)
	12224.6 A.
(QI-12)
	[image: image47.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	12
	MBAR3346
	12210 A.
(D2-L)
	12236 A.
(D1-U)
	12818 A.
(D2-U)
	[image: image48.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	13
	MBBL2185
	9550 A.
(D1-L)
	11487 A.
(D1-L)
	12220 A.
(D2-L)
	12429 A.
(D1-L)
	12615 A.
(D1-L)
	12645 A.
(D1-L)
	12750 A.
(D1-L)
	12840 A.
(D1-U)
	12900 A.
[image: image49.jpg]



	-
	-
	-
	-

	14
	MBAR3327
	9935.7 A.
(D2-L)
	12900 A.
[image: image50.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	15
	MBAL3147
	12616 A.
(D2-U)
	12850 A.
[image: image51.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	16
	MBAR3351
	12461.3 A.
(D2_L)
	12686 A.
(D1-L)
	12850 A.
(D2-U)
	[image: image52.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	17
	MBAR3175
	12457 A.
(D2-U)
	12850 A.
[image: image53.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-


Back to Homepage | For print 

Week 23  :  29 May to 04 June 05  (17 Magnet Tests)
[3147 had ROBTC ]
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END May05 STATISTICS & PERFORMANCE CURVES
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	Date
	Bench
	Problem
	Type
	Application/

soft
	Action/advise

	Tuesday _31_A
	TBD2


	SPA on RMD when going to stand by mode
	AB/PO


	
	Memory fault cleared by Federico. Reset of the PC14 kA before launching the test.

	Tuesday _31_N
	TBE1


	Offset on Potaim card Vsupra_ext not stable + VClamp CD triggering 


	MTM


	
	Federico solved many problems:

· 1 potaim card is out of order,

· several Vtaps missing at the input of potaim crate nr.2 i.e. clamp VT1 to VT4

	Thursday_02_A
	TBA2


	T_sensor in the MB3344 is not showing the good slope although there is a perfect straight line.

	MTM


	
	Federico called. He explained that the calibration coefficients for the interpolation were changed for the TT821 and TT148 both for in AQA and IAP

	Thursday_02_A
	TBD1


	AC compensation on SSS202 could not be carried out.. Trigger is on Potaim cards Vbus_Ext & D1-D2_lg.
	MTM


	
	Solved using the adaptation.

	Friday_03_N
	TBE2


	High current indicator on the CFB is not flashing
	MTM


	
	

	Saturday_04_M
	TBD1


	During the generation of the “ub file” for PT8, DAQ received a trigger “ read  out starts”, system switched off.

Same message received when the system is switched off. 
	AB/CO?


	
	E-mail sent to Giorgio.

	Saturday_04_M
	TBD2


	Polarity check on SSS613.

Voltage applied of. 10 V, current indicated less than 1 A.
	MTM


	
	Giorgio called. He advised to apply the max Voltage of 40 V and to see if the current increased proportionally.

He suggested to do the test as for the arc SSS. In any case there is no diode for this SSS.

	Saturday_04_M
	TBC1


	HV test after cold test using HV rack XMT MY92 on MB3295.Wrong voltage applied in the first step: 700V instead of 600 V. Test was stopped with a comment "Applied voltage exceeded by more than 10% then specified" Run recovery didn't worked.

Also it was found that the C11 connector was damaged for this 
	MTM


	
	Rack was changed and test proceed smoothly?

Test failed because cryo-heater connector was not working. 

Giorgio advise to go on with the test, and to open NC to have the connector repaired later on.

	Saturday_04_N
	TBB1


	PLCSE blocked.
	MTM


	
	Switched off/on the PLCSE.

	Saturday_04_N
	TBB1


	De-excitation test PT8.2 for SSS. The PC did not follow the 350A/s ramp rate specified by the test.
	MTM


	
	The de-ramp happens at ~120A/s.

	Sunday_05_M
	All


	MTF not working since mid night?
	IT?


	
	Restart to work  at 8..30 Am

	Sunday_05_M
	TBF2


	MM1 on MB3134. Test cannot be launched.

Magnets setting file had no Temperature data for flat measurements
	MTM


	MMP
	Ramp rate was put at 0A/s. Changed with 10A/s and test can be launched..

Giorgio corrected it, then TT821 is OK

	Sunday_05_A
	All


	Quench during MM2 (LHC cycle) for MB3134.
	ACR?
	
	Giorgio contacted.

Possible explanation :

· too long time at 11850 A: Tleads B too high and SPA occurred

· quench during the de-ramp.

	Monday_06_M
	TBE2


	Error in Prep 2.2 (QH discharge HF); tminV for H4 is 1624.
	IT?


	
	Maryline checked  and advised to continue.











